Author's response to reviews

Title: Aqueous extract of post-fermented tea reverts the hepatic steatosis of hyperlipidemia rats by regulating the lipogenic genes expression and hepatic fatty acid composition

Authors:

Liang Zhang (zhli2091@sina.com)
Jie Zhou (jz_ahau@sina.com)
Jingsong Zhang (jsz_ahau@sina.com)
Xiaochun Wan (xcwan_ahau@sina.com)

Version: 2 Date: 7 April 2014

Author's response to reviews:

1. We made a comprehensive re-check on the English grammar of this manuscript. We also invited a American scientist of Food science department of UMASS Amherst to review the whole manuscript carefully. We made a great progress on written English.

2. This study was approved by the Institutional Animal Care and Use Committee of Anhui agricultural university and carried out according to the guidelines of Care and Use of Animals Laboratory.

3. The plant species of the plant material was formally identified by Professor Xiaochun Wan, who is a famous scientist of tea research.